Author's response to reviews

Title: A theoretical framework to describe communication processes during medical disability assessment interviews

Authors:

H Jolanda van Rijssen (j.vanrijssen@vumc.nl)
Antonius J M Schellart (ton.schellart@vumc.nl)
Johannes R Anema (h.anema@vumc.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)

Version: 3 Date: 24 September 2009

Author's response to reviews: see over
Title: A theoretical framework to describe communication processes during medical disability assessment interviews

Authors: H Jolanda van Rijssen (j.vanrijssen@vumc.nl)
Antonius J M Schellart (ton.schellart@vumc.nl)
Johannes R Anema (h.anema@vumc.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)

Version: 2 Date: 7 September 2009

Response: 24 September 2009
RE: Formatting Changes for MS: 2137508507274921

Dear BioMed Central Editorial Production Team,

At your request we have made the following formatting changes in our manuscript:
  Major revisions:
  - The lay-out of Table 1 has been changed. All vertical elements and merged cells that cross rows have been removed.
  - Reference 23 has been cited in the text (page 8).
  Minor revisions:
  - All highlighting and tracking have been removed from the manuscript.
  - Table and figure placement text has been removed.
  Final corrections:
  - Errors in Figure 1 have been corrected.

Sincerely,
On behalf of all authors,

Allard J. van der Beek (corresponding author)
VU University Medical Centre, EMGO Institute for Health and Care Research
Department of Public and Occupational Health
Email: a.vanderbeek@vumc.nl